Author’s response to reviews

Title: Cardiac tamponade and para-aortic hematoma post elective surgical myocardial revascularization on a beating heart - a possible complication of the Lima-stitch.

Authors:

Anna Marcinkiewicz (annamar87@o2.pl)
Ryszard Jaszewski (ryszard.jaszewski@umed.lodz.pl)
Katarzyna Piestrzeniewicz (kpiestrzeniewicz@o2.pl)
Radoslaw Zwolinski (radekzwolinski@poczta.onet.pl)

Version: 4  Date: 18 March 2014

Author’s response to reviews:

Lodz, 18.03.2014

Dear Editors,

The authors read the manuscript once more, checked the spelling and made some copy-editing corrections, when mandatory (not marked in the text). The spelling within the section Authors’ Contributions was corrected. The numbering was deleted from the Consent, Authors’ Contributions and Acknowledgments sections. The sentence about ethical approval was removed as unnecessary. The title Conflict of Interests was given to the authors declaration about not having competing interests. These revisions were highlighted in the text.

We hope that our revision is satisfactory.

Your sincerely

Anna Marcinkiewicz MD on behalf of myself and the co-authors.